Author's response to reviews

Title: Good continuum of HIV care in Belgium despite weaknesses in retention and linkage to care among migrants

Authors:

Dominique Van Beckhoven (dvanbeckhoven@wiv-isp.be)
Eric Florence (eflorence@itg.be)
Jean Ruelle (jean.ruelle@uclouvain.be)
Jessika Deblonde (jdeblonde@wiv-isp.be)
Chris Verhofstede (chris.verhofstede@UGent.be)
Steven Callens (STEVENCALLENS@UGent.be)
Ellen Vancutsem (Ellen.Vancutsem@uzbrussel.be)
Patrick Lacor (Patrick.Lacor@uzbrussel.be)
Rémy Demeester (rdemeest@ulb.ac.be)
Jean-Christophe Goffard (Jcgoffar@ulb.ac.be)
André Sasse (asasse@wiv-isp.be)

Version: 2
Date: 31 March 2015

Author's response to reviews:

Dear editor,

We are resubmitting the manuscript « Good continuum of HIV care in Belgium despite weaknesses in retention and linkage to care among migrants”, initially submitted on the Thursday 26th of march to the BMC Infectious Diseases. We have modified the manuscript according to your requirements.

Line numbering has been added in the manuscript.

Further information on the ethical approval received for the HIV/AIDS surveillance and the submission and authorization to the privacy commission have been added in the Methods section of the manuscript. We have also added that our institute is also legally entitled to HIV/AIDS surveillance by a royal decree.

A section with the list of abbreviations used in the text has been added in the manuscript.

The email address of all co-authors have been added in the title page of the manuscript.

I hope that these modifications will fulfil your requirements.
Yours sincerely,

Dominique Van Beckhoven
(corresponding author)